Author's response to reviews

Title: Exercise therapy, manual therapy, or both, for osteoarthritis of the hip or knee: a factorial randomised controlled trial protocol

Authors:

J H Abbott (haxby.abbott@otago.ac.nz)
M C Robertson (clare.robertson@otago.ac.nz)
Joanne E McKenzie (jo.mckenzie@stonebow.otago.ac.nz)
G D Baxter (david.baxter@otago.ac.nz)
Jean-Claude Theis (jean-claude.theis@stonebow.otago.ac.nz)
A J Campbell (john.campbell@stonebow.otago.ac.nz)
the MOA Trial team (haxby.abbott@otago.ac.nz)

Version: 2 Date: 17 December 2008

Author's response to reviews: